Author's response to reviews

Title: Ligation-Assisted Endoscopic Enucleation for the diagnosis and resection of Small Gastrointestinal Tumors Originating from the Muscularis Propria

Authors:

Jintao Guo (guojt@sj-hospital.org)
Zhijun Liu (liuzj1@sj-hospital.org)
Siyu Sun (sunsivucmu@yahoo.com.cn)
Sheng Wang (yeji19830227b@yahoo.com.cn)
Nan Ge (yuefan_0225@yahoo.com.cn)
Guoxin Wang (wanggxcmu@yahoo.com.cn)
Xiang Liu (liuxiangcmu@yahoo.com.cn)
Xianghong Yang (yangxianghongcmu@yahoo.com.cn)

Version: 2 Date: 23 December 2012

Author's response to reviews: see over
Dear Mr. Manibo,

Merry Christmas and Happy New Year! Best wishes to you and your family! We also would like to take this opportunity to say thank you for your advices and suggestions.

We have made Point-by-point changes according to your advices and uploaded the revised version, Thanks you so much!

**Editorial Requirements:**

**Request for Competing Interest section.**

Response: Thank you, we placed the declaration of competing interests after the Conclusion/Abbreviations in the revised manuscript. (Page 7)

**Request for Authors Contribution section.**

Response: Thank you, we placed the Authors’ Contributions section after the Competing Interest section in the revised manuscript. (Page 7)

**Request for Acknowledgment section.**

Response: Thank you, we placed Acknowledgment section after the Authors contribution section in the revised manuscript. (Page 7)

**Please note that table 1 contains identifying information (age, sex, etc) – could you clarify in the revised manuscript that you received consent from each patient to publish this specific information or merge this information so that individuals cannot be identified.**
Response: Thank you, we have received consent from each patient to publish this specific information, which was clarified in the revised manuscript (Page 11: just below the table)

**Please to remove dates, etc in your figures.

Response: Thank you, we removed dates, etc in our revised figures (Figure 2).

Thanks you so much, we would be grateful and make changes as soon as possible if you give further suggestions and advises.

Best Wishes to you and your family!

Thanks you so much!

Yours sincerely,

Si-yu Sun

MS: 3379149498716856
Research article
Ligation-Assisted Endoscopic Enucleation for the diagnosis and resection of Small Gastrointestinal Tumors Originating from the Muscularis Propria
Jintao Guo, Zhijun Liu, Siyu Sun, Sheng Wang, Nan Ge, Guoxin Wang, Xiang Liu and Xianghong Yang
BMC Gastroenterology (Section: Gastroesophageal disorders)